Author's response to reviews

Title: Clinical characteristics and prognostic factors of bone lymphoma: A single-center experience with focus on the clinical significance of multifocal bone involvement in primary bone lymphoma

Authors:

Huanwen Wu (whw14093@163.com)
Marilyn M. Bui (Marilyn.Bui@moffitt.org)
Haipeng Shao (Haipeng.Shao@moffitt.org)
Lubomir Sokol (Lubomir.Sokol@moffitt.org)
Eduardo M. Sotomayor (eduardo.sotomayor@moffitt.org)
Ling Zhang (Ling.Zhang@moffitt.org)

Version: 2
Date: 2 April 2014

Author's response to reviews:

Dear editor,

Line and page numbers have been included, and all page breaks have been deleted in our manuscript. Thanks.

Huanwen Wu